Author's response to reviews

Title: Assessing secondary attack rates among household contacts at the beginning of the influenza A (H1N1) pandemic in Ontario, Canada, April-June 2009: A prospective, observational study.

Authors:

Rachel Savage (rachel.savage@oahpp.ca)
Michael Whelan (Michael.Whelan@ontario.ca)
Ian Johnson (ian.johnson@oahpp.ca)
Elizabeth Rea (erea@toronto.ca)
Marie LaFreniere (marie.lafreniere@peelregion.ca)
Laura C Rosella (laura.rosella@oahpp.ca)
Freda Lam (freda.lam@oahpp.ca)
Tina Badiani (Tina.Badiani@ontario.ca)
Anne-Luise Winter (Anne-Luise.Winter@oahpp.ca)
Deborah J Carr (dcarr@county.oxford.on.ca)
Crystal Frenette (crystal.frenette@peelregion.ca)
Maureen Horn (maureen.horn@peelregion.ca)
Kathleen Dooling (kathleen.dooling@peelregion.ca)
Monali Varia (monali.varia@peelregion.ca)
Anne-Marie Holt (aholt@hkpr.on.ca)
Vidya Sunil (vsunil@hkpr.on.ca)
Catherine Griff (catherine.griff@mlhu.on.ca)
Eleanor Paget (eleanor.paget@mlhu.on.ca)
Michael King (kingm@sdhu.com)
John Barbaro (John.Barbaro@smdhu.org)
Natasha S Crowcroft (natasha.crowcroft@oahpp.ca)

Version: 5 Date: 1 April 2011

Author's response to reviews:

The only change made to this final manuscript was on page 23, Table 1. For the characteristic "Under 16 yrs", the proportion listed for secondary cases was 50.0 when it should have been 49.0. This has now been corrected. No other changes were made.